

# 

**Citation:** Suárez M, Valencia BM, Jara M, Alba M, Boggild AK, Dujardin J-C, et al. (2015) Quantification of *Leishmania* (*Viannia*) Kinetoplast DNA in Ulcers of Cutaneous Leishmaniasis Reveals Inter-site and Inter-sampling Variability in Parasite Load. PLoS Negl Trop Dis 9(7): e0003936. doi:10.1371/journal. pntd.0003936

**Editor:** Alain Debrabant, US Food and Drug Administration, UNITED STATES

Received: March 15, 2015

Accepted: June 26, 2015

Published: July 23, 2015

**Copyright:** © 2015 Suárez et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

**Funding:** This work was financially supported by an Internal Competitive Fund of Universidad Peruana Cayetano Heredia ("Fondo Concursable para Apoyo a la Investigación en la UPCH—2011") for covering working costs; and the Belgian Directorate General for Development Cooperation (FA3 project 95502) for personnel (MS, MJ, MA, VA) and facility support at the Molecular Epidemiology Unit of the Instituto de Medicina Tropical Alexander von Humboldt. The RESEARCH ARTICLE

# Quantification of *Leishmania* (*Viannia*) Kinetoplast DNA in Ulcers of Cutaneous Leishmaniasis Reveals Inter-site and Intersampling Variability in Parasite Load

Milagros Suárez<sup>1</sup>, Braulio M. Valencia<sup>1</sup>, Marlene Jara<sup>1</sup>, Milena Alba<sup>1</sup>, Andrea K. Boggild<sup>2,3,4</sup>, Jean-Claude Dujardin<sup>5,6</sup>, Alejandro Llanos-Cuentas<sup>1</sup>, Jorge Arevalo<sup>1,7</sup>, Vanessa Adaui<sup>1,7</sup>\*

 Instituto de Medicina Tropical Alexander von Humboldt, Universidad Peruana Cayetano Heredia, Lima, Peru, 2 Public Health Ontario Laboratories, Public Health Ontario, Toronto, Canada, 3 Department of Medicine, University of Toronto, Toronto, Canada, 4 Tropical Disease Unit, University Health Network, Toronto General Hospital, Toronto, Canada, 5 Department of Biomedical Sciences, Institute of Tropical Medicine, Antwerp, Belgium, 6 Department of Biomedical Sciences, University of Antwerp, Antwerp, Belgium, 7 Departamento de Ciencias Celulares y Moleculares, Facultad de Ciencias y Filosofía, Universidad Peruana Cayetano Heredia, Lima, Peru

\* vanessa.adaui@upch.pe

# Abstract

## Background

Cutaneous leishmaniasis (CL) is a skin disease caused by the protozoan parasite *Leishmania*. Few studies have assessed the influence of the sample collection site within the ulcer and the sampling method on the sensitivity of parasitological and molecular diagnostic techniques for CL. Sensitivity of the technique can be dependent upon the load and distribution of *Leishmania* amastigotes in the lesion.

## Methodology/Principal Findings

We applied a quantitative real-time PCR (qPCR) assay for *Leishmania* (*Viannia*) minicircle kinetoplast DNA (kDNA) detection and parasite load quantification in biopsy and scraping samples obtained from 3 sites within each ulcer (border, base, and center) as well as in cytology brush specimens taken from the ulcer base and center. A total of 248 lesion samples from 31 patients with laboratory confirmed CL of recent onset ( $\leq$ 3 months) were evaluated. The kDNA-qPCR detected *Leishmania* DNA in 97.6% (242/248) of the examined samples. Median parasite loads were significantly higher in the ulcer base and center than in the border in biopsies (*P*<0.0001) and scrapings (*P* = 0.0002). There was no significant difference in parasite load between the ulcer base and center (*P* = 0.80, 0.43, and 0.07 for biopsy, scraping, and cytology brush specimens, respectively). The parasite load varied significantly by sampling method: in the ulcer base and center, the descending order for the parasite load levels in samples was: cytology brushes, scrapings, and biopsies (*P*<0.0001); in the ulcer border, scrapings had higher parasite load than biopsies (*P*<0.0001). There

funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing Interests:** The authors have declared that no competing interests exist.

was no difference in parasite load according to *L*. *braziliensis* and *L*. *peruviana* infections (P = 0.4).

#### **Conclusion/Significance**

Our results suggest an uneven distribution of *Leishmania* amastigotes in acute CL ulcers, with higher parasite loads in the ulcer base and center, which has implications for bedside collection of diagnostic specimens. The use of scrapings and cytology brushes is recommended instead of the more invasive biopsy.

# Author Summary

Cutaneous leishmaniasis (CL) is a parasitic disease of the skin caused by obligate intramacrophage protozoa of the genus *Leishmania* which usually presents as ulcerative lesions at the site of infection. Traditionally, histopathological and diagnostic studies on CL have employed samples collected from the border of the ulcer since this area is believed to contain the highest amount of parasites. However, no formal demonstration of the distribution of *Leishmania* parasites in the ulcer has been provided yet. Focusing on human skin lesions of recent onset ( $\leq$ 3 months) caused by *L*. (*Viannia*) species, we estimated the parasite loads among different skin lesion sites by means of quantitative real-time PCR targeting the parasite kinetoplast DNA. Paired lesion samples collected by use of different sampling methods were analyzed. We found that the ulcerated zone of the lesion contained a higher parasite load than the ulcer border, and that scraping and cytology brush specimens presented higher parasite loads as compared to the more invasive biopsy. Our results have implications for bedside collection of diagnostic and post-therapeutic followup specimens from CL patients.

## Introduction

Cutaneous leishmaniasis (CL) is a parasitic disease of significant public health problem in at least 18 countries of Latin America; about 67,000 CL cases were reported to occur annually in the last decade [1]. The disease is caused by protozoan parasites of the subgenera *Leishmania* (*Viannia*) and *L*. (*Leishmania*), with the former being responsible for most cases. The clinical phenotypes of CL are diverse and range from a single or few cutaneous ulcerative lesions at the site of infection that may heal spontaneously, diffuse and disseminated CL with multiple non-ulcerative lesions, to disfiguring mucocutaneous leishmaniasis that can be life-threatening [2,3]. The severity and outcome of the disease are dependent among others on the immune responses evoked by the host and the infecting *Leishmania* species [4,5].

Parasitological diagnosis of CL relies on either the microscopic demonstration of *Leishmania* amastigotes in cutaneous tissue or the isolation of parasites from lesions in culture [6-8]. While these techniques are highly specific for diagnosing CL, they are insufficiently sensitive [9]. Polymerase chain reaction (PCR)-based testing of skin lesion specimens has become an important tool to diagnose CL, because of its high sensitivity and specificity (up to 100%) [10-12]. Significant progress has been made towards evaluating molecular-based non-invasive methods for the diagnosis of CL that overcome the disadvantages of the traditional, invasive sampling methods such as punch biopsies, aspirates or skin slits/scrapings [13-16]. One such

non-invasive method, cytology brush PCR, has shown potential for widespread use, both in the clinic [15] and field settings [17].

Several studies indicate that the sensitivity of diagnostic methods for CL can be dependent upon the number and dispersion of parasites in the lesion, the method used to sample ulcers, the stage (chronicity) of the lesion, and the technical skills of the personnel [6,9,10,11,18,19]. Conventionally, in accordance to guidelines established by the World Health Organization (WHO) [20], tissue samples have been obtained from the lesion border, where parasite load and the density of inflammatory mononuclear cells harboring parasites are thought to be higher [21]. Evidences supporting that other sampling sites within lesions could result in comparable or even increased sensitivity of parasite detection by microscopy or PCR have been provided [22,23]. In a recent study using quantitative real-time PCR (qPCR) targeting *Leishmania* 18S rDNA, it has been reported that swab sampling over the ulcer allowed to recover a higher amount of parasite DNA as compared to aspirate samples taken from the lesion border [16]. Whether this indicates differences inherent to the sampling methods or truly reflects a higher parasite load in the ulcerated zone of the lesion [16] needs to be ascertained.

The analysis of the load and distribution of Leishmania parasites within the skin lesions would be important not only for determining the best location within the ulcer to obtain samples for diagnostic purposes, but also for an eventual follow up of a patient's response to treatment [24,25]. We herein applied a standardized qPCR assay targeting minicircle kinetoplast DNA (kDNA) [26] to detect and quantify Leishmania (Viannia) parasites in 3 sites within the CL ulcer (raised border, base, and center). Paired lesion samples were collected by use of different sampling methods: a punch biopsy and a dermal scraping from each of the 3 lesion sites, and a cytology brush from each the base and center of the ulcer. The parasite load levels were compared according to the ulcer site, sampling method, and the infecting *Leishmania* species. We restricted this study to lesions originated from patients with acute CL (<3 months), which characteristically have high parasite load in contrast to lesions from patients with chronic disease (>6 months) [10,26]. This fact enabled detection of *Leishmania* and quantifiable parasite load levels in most clinical specimens. Importantly, early diagnosis is considered a desirable control measure for CL. To our knowledge, this is the first report that quantitatively compares the parasite loads among different skin lesion sites and sampling methods by means of qPCR, thereby providing an insight into the likely distribution of *Leishmania* amastigotes in the ulcer for the L. (Viannia) species present in our sample set. The implications of our results on diagnosis of CL and the prognostic applicability are discussed, as well as how they may relate to the immunopathology of the disease.

#### Methods

#### Ethics statement

This study was conducted according to the principles specified in the Declaration of Helsinki and under local ethical guidelines (Universidad Peruana Cayetano Heredia Institutional Review Board). The study protocol, informed consent and sampling procedures were approved by the Institutional Review Boards of the Hospital Nacional Cayetano Heredia and Universidad Peruana Cayetano Heredia (Lima, Peru) for studies involving human subjects. Written informed consent was obtained from all participants prior to enrolment.

#### Sample size calculation

The sample size was calculated using the G\*Power 3.1 software (release 3.1.9.2; available from: <u>http://www.gpower.hhu.de/</u>) [27] to assess the null hypothesis of no difference in parasite load levels between different sampling sites within the CL ulcer. Assuming a medium effect size of

0.5, a significance level of 5%, and a power of 80%, 35 matched pairs of lesion samples were required to be examined (two-sided, Wilcoxon signed-rank test for matched pairs). For significant results, the effect size was assumed to be 'medium', which means an effect visible to the naked eye. Non-significant results were assumed to have a 'small' effect size. We managed to study 31 paired lesion samples from patients presenting with acute CL.

#### Patients

Patients that attended the Leishmaniasis Clinic at the Instituto de Medicina Tropical Alexander von Humboldt, Hospital Nacional Cayetano Heredia, in Lima, Peru, between January and June 2013 for the examination of skin lesions were invited to participate in the study and evaluated for possible eligibility. Patients were considered for enrolment if they presented with ulcerative skin lesions of recent onset (<3 months of evolution), with elevated and infiltrative borders and a lesion size over 1 cm in diameter; and were able to give written informed consent for the sampling procedures. We included adult patients with laboratory confirmed diagnosis of CL, as defined by a positive result on at least 1 of these 3 tests: direct microscopy on Giemsa-stained lesion smears [7], lesion aspirate microculture [28], and qualitative PCR targeting kDNA minicircles [29] on a biopsy specimen obtained from the ulcer border. This diagnostic PCR includes internal control primers for amplifying the human beta-globin gene as previously described [13]. The intradermal leishmanin skin test (LST), used to assess exposure to Leishmania infection, was performed on CL patients before treatment, as described elsewhere [30,31]. We excluded patients allergic to local anesthetics, with clinical evidence of bacterial or fungal superinfection of the ulcer (when possible), with any contraindication to skin biopsy and those undergoing active treatment for CL. In three cases with secondarily infected ulcers, patients were treated with a 5-day course of antibiotics before sample collection.

#### Lesion sampling

In order to analyze the distribution and load of *Leishmania* amastigotes within the cutaneous lesion, samples were collected from 3 different sites, in the following order: the center of the ulcer, the base (inner border) of the ulcer, and the raised border of the ulcer (Fig 1); using a randomly chosen coordinate defined as North, South, East or West, taking as reference the lateral and longitudinal axes of the human body. If the patient had more than one lesion, the most active and typical indurated ulcer was selected. Eight specimens were collected from a single lesion per patient: a punch biopsy and a dermal scraping from each of the 3 lesion sites, and a cytology brush from each the center and base of the ulcer. The order of sampling was: biopsy, scraping, and cytology brush. All samples were taken by the same physician in order to avoid inter-individual variation.

Prior to sampling, lesions were cleansed with topical antiseptics, removed from any overlying scab or crust with saline solution and anesthetized with 1 cc of lidocaine 1%.

**Biopsies.** A small tissue fragment of 1.5 mm in diameter was obtained from the ulcer center, base and raised border, using a sterile disposable punch (Miltex), at a randomly chosen coordinate within the cutaneous lesion.

**Scrapings.** Lesion material was scraped from the ulcer center, base and from an incision made at the raised border, using a sterile lancet; this was done in the same coordinate but adjacent to the point from where biopsy samples were obtained.

**Cytology brushes.** A sterile cervical cytology brush (Cervisoft, Puritan Medical Products) was rolled clockwise at a single point of the ulcer center and base 5 times each in sequence in order to collect lesion cellular and exudative material, as described by Valencia et al. [15]; this





doi:10.1371/journal.pntd.0003936.g001

was done in the same coordinate but adjacent to the point from where scrapings were obtained.

Clinical specimens were stored at  $-20^{\circ}$ C in a 1.5 mL Eppendorf tube containing 700  $\mu$ L 100% ethanol for subsequent molecular analysis.

#### Isolation of DNA from biopsies, lancets and cytology brushes

Prior to DNA extraction, samples were centrifuged at 8000 g for 2 min and ethanol was discarded. Biopsied tissue was disaggregated with a sterile scalpel. Disaggregated tissue, lancets and cytology brushes were subjected to overnight lysis with Proteinase K and processed for DNA isolation using a column-based method (High Pure PCR template preparation kit, Roche), according to the manufacturer's instructions. The isolated DNA was then quantified by fluorometry using the Quant–iT Broad Range dsDNA Assay kit (for biopsies) and the Quant-iT High Sensitivity dsDNA Assay kit (for scrapings and cytology brushes) on the Qubit fluorometer (Invitrogen). DNA samples were diluted to 5 ng/µL; those samples below this concentration were added directly into the PCR reaction.

#### Parasite species identification

Parasites were typed using the heat-shock protein 70 gene (*hsp70*) PCR-N variant followed by restriction fragment length polymorphism (RFLP) analysis using the restriction enzymes *BsaJI* and *RsaI* as in Montalvo et al. [33].

#### Detection and quantification of Leishmania (Viannia) spp.

We applied a SYBR Green-based qPCR assay targeting kDNA minicircles to detect and quantify *Leishmania* (*Viannia*) parasites in clinical samples, as previously described [26]. Each kDNA-qPCR run included a standard curve of *L*. (*V*.) *braziliensis* (MHOM/BR/75/M2904) DNA ranging from  $5 \times 10^4$  to  $5 \times 10^{-3}$  parasite DNA equivalents/reaction (run in duplicate); a positive control with known amount of *Leishmania* parasites, which consisted of a mix of *Leishmania* DNA and human genomic DNA in order to mimic clinical specimens (run in triplicate); a negative control (human genomic DNA from peripheral blood mononuclear cells of a healthy donor; run in triplicate); and a blank (no-template control; run in triplicate). The standard curves (inter-assay reproducibility, n = 11) showed a mean square error (MSE) of  $\leq 0.111$ , correlation coefficient ( $r^2$ ) of  $\geq 0.998$  and slopes of 3.28 (mean)  $\pm 0.05$  (standard deviation), indicating a high amplification efficiency ( $\geq 1.99$ ) (2 would indicate 100% PCR efficiency). The positive control showed a mean of 7,640 parasites and an inter-assay coefficient of variation of 7.8% (n = 11 independent runs). All clinical samples were run in duplicate; if replicates differed by a standard deviation of >0.35 in Cq (quantification cycle) values (>0.5 cycles), they were retested.

A sample was quantified when it had a Cq value falling within the range of the standard curve. The highest dilution of template of the standard curve was defined as the lower limit of quantification (LOQ). Samples with *Leishmania* DNA levels below the LOQ could be detected; they were considered positive (qualitative detection) only if their melting curves had the same profile as those of the standards included in the same experiment. The *Leishmania* parasite load was calculated as follows: [parasite DNA equivalents per reaction/amount of tissue DNA per reaction]  $\times 10^3$ , expressed as the number of *Leishmania* parasites per µg of tissue DNA.

#### Statistical analysis

Frequencies and proportions were used to describe categorical variables while median and interquartile range or mean and standard deviation were used for numeric continuous variables.

To assess whether the median parasite load in clinical specimens differed significantly according to the skin lesion site or the sampling method, analyses for paired samples using Friedman (with Dunn's post-hoc test) and Wilcoxon signed rank tests were performed. The correlation degree between the parasite load measurements in scraping and cytology brush specimens with respect to those in biopsy specimens was calculated using the Spearman's rank correlation test. The association between the *Leishmania* load and the parasite species was evaluated using the Mann-Whitney U test.

Statistical analyses were performed under a 5% significance level, using the GraphPad Prism v5.02 software.

#### Results

#### Study population

Demographic, epidemiological, and clinical characteristics of patients are summarized in <u>S1</u> <u>Table</u>. Thirty-one patients with laboratory confirmed CL were enrolled: 29 (93.6%) men and 2 (6.5%) women, with median age of 34 years (range 19–75 years) and median disease duration of 2 months (range 1–3 months). Median number of lesions was 1 (range 1–10), with 21 patients (67.7%) presenting with single lesions and 9 patients (29%) presenting with multiple lesions. Bacterial superinfection was present in only 3 (9.7%) lesions. Twenty-eight patients (90.3%) had a first episode of CL and only one patient (3.2%) had a reinfection. Median duration of exposure in the risk area (i.e. stay in area of endemicity) was 3 months (range 1.5 days–75 years).

## Positivity of the kDNA-qPCR assay and diagnostic tests

The kDNA-qPCR assay detected *Leishmania* DNA in 97.6% (242/248) of the examined lesion specimens. The overall qPCR positivity per lesion-analysis taking into account the 3 lesion sites and sampling methods (96.8%; 95% CI: 74.3–100.0%) was higher than that of smear microscopy (80.6%; 95% CI: 62.5–92.6%), microculture (88.5%; 95% CI: 69.9–97.6%), and LST (72.4%; 95% CI: 52.8–87.3%), whereas it was comparable to that of the qualitative kDNA PCR (96.8%; 95% CI: 83.3–99.9%).

# Positivity of the kDNA-qPCR assay according to the ulcer site and sampling method

The performance of the qPCR for *Leishmania* DNA detection (no quantification at this stage) was assessed in the 3 lesion sites and sampling methods. In the ulcer border, *Leishmania* DNA was detected by qPCR in 100% (31/31) of the scrapings and in 90% (28/31) of the biopsies. In the ulcer base, 100% (31/31) of the biopsy specimens, 97% (30/31) of the scraping specimens, and 97% (30/31) of the cytology brush specimens tested positive for *Leishmania* DNA. In the ulcer center, *Leishmania* DNA was detected in 100% (31/31) of the examined biopsies, in 100% (31/31) of the scrapings, and in 97% (30/31) of the scrapings.

# Comparison of parasite loads according to the ulcer site and sampling method

The kDNA-qPCR assay allowed the quantification of the parasite load in 238 out of 248 lesion specimens (96%). As for the 10 specimens that could not be quantified, 4 corresponded to dermal scrapings with detectable but not quantifiable parasite load, whereas 6 specimens among biopsies, scrapings and cytology brushes were qPCR negative. These 10 specimens corresponded to 3 patients. After exclusion of those 3 patients from the analysis, the quantified parasite load assessment across ulcer sites and sampling methods. The parasite load (PL) in skin lesion specimens varied from  $2.53 \times 10^1$  to  $5.72 \times 10^6$  parasites per µg of tissue DNA. The PL levels per skin lesion site and sampling method are given in Table 1 and depicted graphically in Fig 2.

**Parasite loads according to the ulcer site.** Median PL differed among the 3 sites of the ulcer in biopsies and scrapings (P<0.0001 and P = 0.0002, respectively, Friedman test), with specimens from the ulcer base and center having significantly higher PL than those from the ulcer border (Dunn's post-hoc test). There was no significant difference in PL between the ulcer base and center, either in biopsy specimens (P = 0.80, Wilcoxon signed rank test), scraping specimens (P = 0.43, Wilcoxon signed rank test), or in cytology brush specimens (P = 0.07, Wilcoxon signed rank test) (Table 1).

**Parasite loads according to the sampling method.** Median PL differed by sampling method: in both the ulcer base and center, (i) a higher PL could be quantified from cytology brushes and dermal scrapings as compared to biopsies; and (ii) cytology brushes showed a higher PL than scrapings (P<0.0001, Friedman test with Dunn's post-hoc test). In the ulcer border, dermal scrapings contained a higher PL than biopsies (P<0.0001, Wilcoxon signed rank test) (Table 1). Parasite load measurements on biopsies vs. scrapings or cytology brushes were highly correlated in all lesion sites (Spearman's rho range 0.75–0.93; P<0.0001, S2 Table), indicating that the parasite load trend was consistent across sampling methods.

Skin lesion site	Sampling method	Parasite load <sup>‡</sup>		
		Median	IQR	Range
Raised border <sup>¶</sup>	Biopsy <sup>¥</sup>	2.96 × 10 <sup>3</sup>	$1.59 \times 10^{2}$ - $1.94 \times 10^{4}$	$2.87 \times 10^{1-2.88} \times 10^{5}$
	Scraping <sup>†</sup>	$5.96 \times 10^{4}$	$3.14 \times 10^{3}$ -1.31 × 10 <sup>5</sup>	1.81 × 10 <sup>2−</sup> 6.84 × 10 <sup>5</sup>
Base (inner border)*	Biopsy <sup>¥</sup>	$3.33 \times 10^{4}$	$1.86 \times 10^{3-}1.32 \times 10^{5}$	$7.44 \times 10^{1-}1.56 \times 10^{6}$
	Scraping <sup>†</sup>	$7.61 \times 10^4$	$7.29 \times 10^{3-}3.90 \times 10^{5}$	2.24 × 10 <sup>2–</sup> 1.17 × 10 <sup>6</sup>
	Cytology brush <sup>§</sup>	$1.76 \times 10^{5}$	$1.76 \times 10^{4-}9.67 \times 10^{5}$	$1.14 \times 10^{3-4.12} \times 10^{6}$
Center*	Biopsy <sup>¥</sup>	$2.11 \times 10^{4}$	$2.14 \times 10^{3-1}$ .19 $\times 10^{5}$	$2.53 \times 10^{1-2.29} \times 10^{6}$
	Scraping <sup>†</sup>	1.11 × 10 <sup>5</sup>	$1.50 \times 10^{4-}3.78 \times 10^{5}$	1.72 × 10 <sup>2–</sup> 2.92 × 10 <sup>6</sup>
	Cytology brush <sup>§</sup>	$1.61 \times 10^{5}$	$2.69 \times 10^{4-1.04} \times 10^{6}$	$1.07 \times 10^{3-5.72} \times 10^{6}$

#### Table 1. Leishmania parasite load levels per skin lesion site and sampling method.

Note. IQR, interquartile range (25<sup>th</sup> percentile–75<sup>th</sup> percentile).

Data shown are the quantified paired parasite load results (8 specimens per lesion) corresponding to 28 patients.

<sup>‡</sup>Number of parasites per µg of tissue DNA.

<sup>¶</sup>P<0.0001, for the comparison of parasite loads between biopsy and scraping specimens (Wilcoxon signed rank test).

\*P<0.0001, for the comparison of parasite loads among biopsy, scraping, and cytology brush specimens (Friedman test with Dunn's post hoc test).

\*P<0.0001, for the comparison of parasite loads among biopsy specimens of the ulcer border, base, and center (Friedman test with Dunn's post hoc test). †P = 0.0002, for the comparison of parasite loads among scraping specimens of the ulcer border, base, and center (Friedman test with Dunn's post hoc test).

<sup>§</sup>P = 0.07, for the comparison of parasite loads between cytology brush specimens of the ulcer base and center (Wilcoxon signed rank test).

doi:10.1371/journal.pntd.0003936.t001

## Comparison of parasite loads according to the infecting species

Causative species was identified in 29 of 31 (93.5%) patients having lesion specimens with sufficient amplifiable DNA: 20 patients were infected with *L*. (*V*.) *braziliensis*, 7 with *L*. (*V*.) *peruviana*, 1 with *L*. (*V*.) *guyanensis*, and 1 with *L*. (*V*.) *lainsoni*. There was no significant difference in PL according to the infecting species, taking into account in this analysis only the most well



**Fig 2.** Parasite load levels in clinical samples according to skin lesion site. Data shown are the quantified paired parasite load results (8 specimens per lesion) corresponding to 28 patients. (A) Comparison of biopsy specimens taken from the ulcer border, base, and center (P<0.0001, Friedman test with Dunn's post hoc test). (B) Comparison of dermal scraping specimens taken from the ulcer border, base, and center (P = 0.0002, Friedman test with Dunn's post hoc test). The asterisks shown in A and B indicate statistically significant differences between corresponding groups according to Dunn's post hoc test. (C) Comparison of cytology brush specimens taken from the ulcer base and center (P = 0.07, Wilcoxon signed rank test).

doi:10.1371/journal.pntd.0003936.g002

Species	Number	Parasite load <sup>‡</sup>			
		Median*	IQR	Range	
L.(V.) braziliensis	20	1.47 × 10 <sup>5</sup>	$2.89 \times 10^{4}$ - $3.72 \times 10^{5}$	$6.40 \times 10^{0}$ - 1.17 $\times 10^{6}$	
L.(V.) guyanensis	1	$6.78 \times 10^{3}$	_	_	
L.(V.) peruviana	7	$8.99 \times 10^{4}$	$8.79 \times 10^{3}$ -2.11 × 10 <sup>5</sup>	$2.40 \times 10^{2}$ -6.30 × 10 <sup>5</sup>	
L.(V.) lainsoni	1	3.04 × 10 <sup>3</sup>	_	_	
Unknown <sup>¶</sup>	2	$2.40 \times 10^{1}$	$1.27 \times 10^{1}$ - $3.53 \times 10^{1}$	$1.27 \times 10^{1}$ - $3.53 \times 10^{1}$	

#### Table 2. Causative Leishmania species and pre-treatment parasite load in tissue.

**Note.** IQR, interquartile range (25<sup>th</sup> percentile–75<sup>th</sup> percentile).

<sup>‡</sup>Number of parasites per µg of tissue DNA.

<sup>1</sup>Species identification was not performed on these specimens because of insufficient concentration of amplifiable DNA based on kDNA PCR band thickness (as obtained in the qualitative, diagnostic PCR assay).

\*P = 0.4, by Mann-Whitney U test for L. (V.) braziliensis and L. (V.) peruviana.

doi:10.1371/journal.pntd.0003936.t002

represented species (i.e. *L*. (*V*.) *braziliensis* and *L*. (*V*.) *peruviana*) (P = 0.4, Mann-Whitney U test) (<u>Table 2</u>).

#### Discussion

Here we assessed whether the Leishmania (Viannia) parasite load differs by sampling site within CL ulcers and sampling method by means of qPCR. We observed that a significantly lower amount of parasites was quantified from the ulcer border as compared to the ulcer base and center. The fact that this finding was similarly observed with lesion biopsy and scraping specimens points out its robustness. This finding called our attention because most of available studies in the literature on skin lesions caused by Neotropical Leishmania parasites are based on biopsies collected from the border of the ulcer, in accordance with WHO recommendations [20], as this lesion site is regarded to likely concentrate a greater amount of parasites and viable infected mononuclear phagocytes compared to the necrotic center of the ulcer [21]. Nevertheless, consistent with our findings, in a recent study that evaluated the use of swab sampling over the ulcer coupled to qPCR for diagnosis of CL, Adams et al. [16] found indications of a greater quantity of parasite DNA in the ulcerated zone of the lesion (as compared to that found in the ulcer border using aspirate samples). Furthermore, Ramírez et al. [23] reported a significant increase in the sensitivities of microscopy and conventional PCR of dermal scrapings when samples were collected from the central region of the bottom of the ulcer rather than from the margin of the lesion. This appeared to be related to a higher parasite load and easily detectable amastigotes in that area [23].

The differences in parasite load among sites within CL ulcers revealed herein may be related to the undergoing immunopathological process within the ulcer. Studies of lesion biopsies (taken, where known, from the border of the lesion) from patients infected with *L*. (*Viannia*) parasites have shown an inflammatory infiltrate in the dermis composed mainly of lymphocytes, macrophages, and plasma cells [34–38]. *Leishmania* amastigotes were seen within dermal macrophages located in the papillary dermis [35] and mid-dermis [37]. Notably, Gutierrez et al. [38] found a significant association between necrosis, relative abundance of tissue macrophages, and the presence of amastigotes in lesions of less than 6 months' duration. The ulcerated zone of the lesion is mainly composed of dead cells, as shown by the presence of focal necrosis of the dermis as well as epidermal disruption [34–37]. In contrast, in the raised border adjacent to the ulceration, the epidermis exhibits hyperplasia and thickening [36,38].

Despite the fact that the quantified parasite loads varied widely among examined lesion samples, we found that there was no significant difference in parasite loads with respect to *L*. (*V*.) *braziliensis* and *L*. (*V*.) *peruviana* infections demonstrated in CL lesions, a finding consistent with a previous study [26]. As these 2 *Leishmania* species can lead to different clinical prognoses [39], our data lends support to a lack of association between parasite load and the degree of pathology noted in studies on human [26] and experimental murine [40] CL. Thus, differences in pathogenicity should rely on other aspects of the host-parasite interaction.

Remarkably, compared to biopsies, we found that a higher parasite load could be quantified from cytology brushes and dermal scrapings. There are two possible explanations for this finding. First, the difference in parasite loads quantified amongst sampling methods pointed out that the Leishmania amastigotes would not be homogeneously distributed along the skin compartments (i.e. cross-sectional), with a higher abundance of parasites to be found in the upper layers of the dermis. This is corroborated by other studies that showed that the cell types in the CL lesion infiltrates were non-randomly distributed, with macrophages and parasites being most frequently found in the mid-dermis [37] as well as in the papillary dermis [35]. Second, there are differences inherent to the sampling instrument. Skin punch biopsy is the only method that allows to sample the full-thickness skin; therefore, the ratio of human host DNA to parasite DNA in this diagnostic specimen is several-fold higher compared to scrapings and cytology brushes. This can decrease the sensitivity of detection of the pathogen in clinical samples [19,41,42]. In contrast, scraping and cytology brush sampling is more superficial and allows to recover both cellular material and tissue fluid [15]; these features reduce the proportion of host cells resulting in improved parasite DNA representation relative to the human host DNA.

Our finding that scrapings and cytology brushes outperform the invasive biopsy in terms of the parasite load quantified is particularly important when considering that invasive specimen collection is a traumatic procedure to the patient frequently associated with risks of bleeding and infection; it entails the risk of body fluid exposure to the healthcare worker via needle stick injury, and is difficult to perform in the pediatric population [13,15,43,44]. Furthermore, it is a complex medical procedure performed by trained medical personnel, normally a dermatologist, and is difficult to perform routinely in endemic settings. Cytology brushes offer the advantage over biopsies and scrapings of being non-invasive, easy to perform and well tolerated by the patient [15,44]. This makes them an attractive alternative not only for diagnosis of CL but also for monitoring patients' response to treatment (through assessment of parasite load kinetics). Such an applicability of cytology brush sampling coupled to kDNA-qPCR has been recently evaluated in a cohort of Peruvian patients with mucosal leishmaniasis [45]. From a practical point of view, our data herein also indicate that samples for routine laboratory diagnosis or an eventual post-treatment follow-up of CL patients can be easily and safely obtained from the ulcer base and center by use of less invasive means, thus obviating the need for any skin incision from the lesion border.

The qPCR assay employed herein targets a multicopy conserved region of minicircle kDNA common to *Leishmania* (*Viannia*) species, present at about 10,000 copies per amastigote [46], thereby allowing to quantify the number of parasites present in the ulcer with high sensitivity. The *Leishmania* kDNA levels detected and quantified in the lesion specimens are likely indicative of the presence of viable parasites, since nuclear and kinetoplast *Leishmania* DNA are rapidly degraded following amastigote death [47]. The variability of the number of kDNA minicircle targets [26,48] was not assessed in the lesion specimens examined here. Nonetheless, our data analysis took into account paired samples (those taken from a same lesion of a patient), which allowed a more accurate estimation of parasite load levels in the ulcer across subjects.

In this study, only patients with recent onset CL ( $\leq$ 3 months) were enrolled. That early stage of CL is associated with a positive parasitological diagnosis and a high parasite load; conversely, chronic CL (>6 months) is characterized by a scarcity of parasites in lesions [10,23,26,38]. Thus, further studies covering both acute and chronic stages of CL caused by *L*. (*Viannia*) parasites are needed to confirm and expand our results. Nevertheless, this study is valuable as it is, to our knowledge, the first report that assesses quantitatively whether the *Leishmania* parasite load differs by both site of sample collection within the skin ulcer and sampling method by means of qPCR.

Our results herein are applicable to ulcerative skin lesions, which represent the most frequent form of localized CL in Latin America. In areas where leishmaniasis is endemic, a smaller number of patients present with other types of cutaneous manifestations (nodular, verrucous, plaques, and papular lesions), either as primarily presentation or in addition to the ulcerative lesion [13,49,50]. Future studies assessing the parasite load in these other types of lesions covering different stages of the disease will add to our understanding of a polymorphic skin disease as CL is.

Our data reveal a picture of the CL ulcer being a complex place, where the process of survival of *Leishmania* amastigotes is occurring, with abundant amastigotes in a highly necrotic tissue. Future studies based on morphometric analysis of histopathological sections are needed to establish the in situ location and quantity of parasites in relation to cellular infiltrates in the ulcerated zone of the lesion, and during different stages of the disease. This may further our understanding of the dynamics of infection in human CL due to *L*. (*Viannia*) species.

#### **Supporting Information**

**S1 Table. Demographic, epidemiological and clinical characteristics of the 31 enrolled patients with CL.** Note. Data are number (%) of cases, unless otherwise indicated. *n*, no. of patients; SD, standard deviation; IQR, interquartile range (25<sup>th</sup> percentile–75<sup>th</sup> percentile). (DOCX)

S2 Table. Correlation between parasite load measurements on biopsy specimens vs. scraping or cytology brush specimens. Note. CI, confidence interval. (DOCX)

#### Acknowledgments

We gratefully acknowledge the patients who participated in this study for their cooperation. We thank Mirko Zimic (Unidad de Bioinformática, Universidad Peruana Cayetano Heredia) for advice on statistical hypothesis testing. We also acknowledge Puritan Medical Products for kindly donating us the cervical cytology brushes (Cervisoft) used in this study.

#### **Author Contributions**

Conceived and designed the experiments: MS MJ JA VA. Performed the experiments: MS MJ MA. Analyzed the data: MS MJ VA. Contributed reagents/materials/analysis tools: JCD ALC. Wrote the paper: MS VA. Contributed to study design: BMV AKB ALC. Enrolled patients, collected samples and clinical data: BMV ALC. Critically revised the manuscript: BMV MJ AKB JCD ALC JA.

#### References

- Alvar J, Vélez ID, Bern C, Herrero M, Desjeux P, Cano J, et al. Leishmaniasis worldwide and global estimates of its incidence. PLoS One. 2012; 7: e35671. doi: <u>10.1371/journal.pone.0035671</u> PMID: 22693548
- Lumbreras H, Guerra H. Leishmaniasis in Peru. In: Chang KP, Bray R, editors. Leishmaniasis. New York: Elsevier, 1985. pp. 297–311.
- Reithinger R, Dujardin JC, Louzir H, Pirmez C, Alexander B, Brooker S. Cutaneous leishmaniasis. Lancet Infect Dis. 2007; 7: 581–596. PMID: <u>17714672</u>
- 4. Grimaldi G Jr, Tesh RB. Leishmaniases of the New World: current concepts and implications for future research. Clin Microbiol Rev. 1993; 6: 230–250. PMID: <u>8358705</u>
- McMahon-Pratt D, Alexander J. Does the *Leishmania major* paradigm of pathogenesis and protection hold for New World cutaneous leishmaniases or the visceral disease? Immunol Rev. 2004; 201: 206– 224. PMID: <u>15361243</u>
- 6. Herwaldt BL. Leishmaniasis. Lancet. 1999; 354: 1191–1199. PMID: 10513726
- Escobar MA, Martinez F, Scott Smith D, Palma GI. American cutaneous and mucocutaneous leishmaniasis (tegumentary): a diagnostic challenge. Trop Doct. 1992; 22 Suppl 1: 69–78;63–4. PMID: <u>1492379</u>
- Murray HW, Berman JD, Davies CR, Saravia NG. Advances in leishmaniasis. Lancet. 2005; 366: 1561–1577. PMID: <u>16257344</u>
- Weigle KA, de Dávalos M, Heredia P, Molineros R, Saravia NG, D'Alessandro A. Diagnosis of cutaneous and mucocutaneous leishmaniasis in Colombia: a comparison of seven methods. Am J Trop Med Hyg. 1987; 36: 489–496. PMID: <u>2437815</u>
- Weigle KA, Labrada LA, Lozano C, Santrich C, Barker DC. PCR-based diagnosis of acute and chronic cutaneous leishmaniasis caused by *Leishmania (Viannia)*. J Clin Microbiol. 2002; 40: 601–606. PMID: <u>11825977</u>
- Bensoussan E, Nasereddin A, Jonas F, Schnur LF, Jaffe CL. Comparison of PCR assays for diagnosis of cutaneous leishmaniasis. J Clin Microbiol. 2006; 44: 1435–1439. PMID: 16597873
- 12. Reithinger R, Dujardin JC. Molecular diagnosis of leishmaniasis: current status and future applications. J Clin Microbiol. 2007; 45: 21–25. PMID: <u>17093038</u>
- Boggild AK, Valencia BM, Espinosa D, Veland N, Ramos AP, Arevalo J, et al. Detection and species identification of *Leishmania* DNA from filter paper lesion impressions for patients with American cutaneous leishmaniasis. Clin Infect Dis. 2010; 50: e1–6. doi: <u>10.1086/648730</u> PMID: <u>19947858</u>
- Mimori T, Matsumoto T, Calvopiña MH, Gomez EA, Saya H, Katakura K, et al. Usefulness of sampling with cotton swab for PCR-diagnosis of cutaneous leishmaniasis in the New World. Acta Trop. 2002; 81: 197–202. PMID: <u>11835896</u>
- Valencia BM, Veland N, Alba M, Adaui V, Arevalo J, Low DE, et al. Non-invasive cytology brush PCR for the diagnosis and causative species identification of American cutaneous leishmaniasis in Peru. PLoS One. 2012; 7: e49738. doi: 10.1371/journal.pone.0049738 PMID: 23185421
- Adams ER, Gomez MA, Scheske L, Rios R, Marquez R, Cossio A, et al. Sensitive diagnosis of cutaneous leishmaniasis by lesion swab sampling coupled to qPCR. Parasitology. 2014; 141: 1891–1897. doi: 10.1017/S0031182014001280 PMID: 25111885
- 17. Valencia BM, Garcia L, Rojas E, Alba M, Montaño N, Angulo C, et al. PCR-based diagnosis of cutaneous Leishmaniasis in two endemic villages from Peru and Bolivia: an articulated strategy founded on local health promoters and conventional sample shipping. Poster session presented at: 63<sup>rd</sup> Annual Meeting of American Society of Tropical Medicine and Hygiene (ASTMH); 2014 Nov 2–6; New Orleans, Louisiana, USA. (Abstract Book, p. 332, abstract nr. 1095).
- Robinson RJ, Agudelo S, Muskus C, Alzate JF, Berberich C, Barker DC, et al. The method used to sample ulcers influences the diagnosis of cutaneous leishmaniasis. Trans R Soc Trop Med Hyg. 2002; 96 Suppl 1: S169–171. PMID: <u>12055833</u>
- Vergel C, Walker J, Saravia NG. Amplification of human DNA by primers targeted to *Leishmania* kinetoplast DNA and post-genome considerations in the detection of parasites by a polymerase chain reaction. Am J Trop Med Hyg. 2005; 72: 423–429. PMID: 15827280
- World Health Organization. Control of the leishmaniases. Report of a WHO Expert Committee. World Health Organ Tech Rep Ser. 1990; 793: 1–158. PMID: <u>2124015</u>
- Vergel C, Palacios R, Cadena H, Posso CJ, Valderrama L, Perez M, et al. Evidence for *Leishmania* (*Viannia*) parasites in the skin and blood of patients before and after treatment. J Infect Dis. 2006; 194: 503–511. PMID: <u>16845635</u>
- Navin TR, Arana FE, de Mérida AM, Arana BA, Castillo AL, Silvers DN. Cutaneous leishmaniasis in Guatemala: comparison of diagnostic methods. Am J Trop Med Hyg. 1990; 42: 36–42. PMID: 2301704

- Ramírez JR, Agudelo S, Muskus C, Alzate JF, Berberich C, Barker D, et al. Diagnosis of cutaneous leishmaniasis in Colombia: the sampling site within lesions influences the sensitivity of parasitologic diagnosis. J Clin Microbiol. 2000; 38: 3768–3773. PMID: <u>11015400</u>
- 24. van der Meide WF, Peekel I, van Thiel PP, Schallig HD, de Vries HJ, Zeegelaar JE, et al. Treatment assessment by monitoring parasite load in skin biopsies from patients with cutaneous leishmaniasis, using quantitative nucleic acid sequence-based amplification. Clin Exp Dermatol. 2008; 33: 394–399. doi: 10.1111/j.1365-2230.2007.02680.x PMID: 18346182
- 25. Dorlo TP, van Thiel PP, Schoone GJ, Stienstra Y, van Vugt M, Beijnen JH, et al. Dynamics of parasite clearance in cutaneous leishmaniasis patients treated with miltefosine. PLoS Negl Trop Dis. 2011; 5: e1436. doi: 10.1371/journal.pntd.0001436 PMID: 22180803
- Jara M, Adaui V, Valencia BM, Martinez D, Alba M, Castrillon C, et al. Real-time PCR assay for detection and quantification of *Leishmania (Viannia)* organisms in skin and mucosal lesions: exploratory study of parasite load and clinical parameters. J Clin Microbiol. 2013; 51: 1826–1833. doi: <u>10.1128/JCM.</u> 00208-13 PMID: 23554201
- Faul F, Erdfelder E, Lang AG, Buchner A. G\*Power 3: a flexible statistical power analysis program for the social, behavioral, and biomedical sciences. Behav Res Methods. 2007; 39: 175–191. PMID: <u>17695343</u>
- Boggild AK, Miranda-Verastegui C, Espinosa D, Arevalo J, Adaui V, Tulliano G, et al. Evaluation of a microculture method for isolation of *Leishmania* parasites from cutaneous lesions of patients in Peru. J Clin Microbiol. 2007; 45: 3680–3684. PMID: <u>17881557</u>
- Lopez M, Inga R, Cangalaya M, Echevarria J, Llanos-Cuentas A, Orrego C, et al. Diagnosis of *Leishmania* using the polymerase chain reaction: a simplified procedure for field work. Am J Trop Med Hyg. 1993; 49: 348–356. PMID: <u>8396860</u>
- Mimori T, Hashiguchi Y, Kawabata M, Gómez EA, De Coronel VV. The relationship between severity of ulcerated lesions and immune responses in the early stage of cutaneous leishmaniasis in Ecuador. Ann Trop Med Parasitol. 1987; 81: 681–685. PMID: <u>3503644</u>
- Maurer-Cecchini A, Decuypere S, Chappuis F, Alexandrenne C, De Doncker S, Boelaert M, et al. Immunological determinants of clinical outcome in Peruvian patients with tegumentary leishmaniasis treated with pentavalent antimonials. Infect Immun. 2009; 77: 2022–2029. doi: <u>10.1128/IAI.01513-08</u> PMID: <u>19237520</u>
- Zvietcovich F, Castañeda B, Valencia B, Llanos-Cuentas A. A 3D assessment tool for accurate volume measurement for monitoring the evolution of cutaneous leishmaniasis wounds. Conf Proc IEEE Eng Med Biol Soc. 2012; 2012: 2025–2028. doi: 10.1109/EMBC.2012.6346355 PMID: 23366316
- Montalvo AM, Fraga J, Maes I, Dujardin JC, Van der Auwera G. Three new sensitive and specific heatshock protein 70 PCRs for global *Leishmania* species identification. Eur J Clin Microbiol Infect Dis. 2012; 31: 1453–1461. doi: 10.1007/s10096-011-1463-z PMID: 22083340
- Barral A, Jesus AR, Almeida RP, Carvalho EM, Barral-Netto M, Costa JM, et al. Evaluation of T-cell subsets in the lesion infiltrates of human cutaneous and mucocutaneous leishmaniasis. Parasite Immunol. 1987; 9: 487–497. PMID: 2957642
- Sotto MN, Yamashiro-Kanashiro EH, da Matta VL, de Brito T. Cutaneous leishmaniasis of the New World: diagnostic immunopathology and antigen pathways in skin and mucosa. Acta Trop. 1989; 46: 121–130. PMID: 2565073
- Pirmez C, Oliveira-Neto MP, Grimaldi Júnior G, Savino W. Immunopathology of American cutaneous leishmaniasis. Modulation of MHC class II gene products by keratinocytes before and after glucantime therapy. Mem Inst Oswaldo Cruz. 1990; 85: 203–209. PMID: 2128362
- Palma GI, Saravia NG. In situ characterization of the human host response to Leishmania panamensis. Am J Dermatopathol. 1997; 19: 585–590. PMID: <u>9415615</u>
- Gutierrez Y, Salinas GH, Palma G, Valderrama LB, Santrich CV, Saravia NG. Correlation between histopathology, immune response, clinical presentation, and evolution in *Leishmania braziliensis* infection. Am J Trop Med Hyg. 1991; 45: 281–289. PMID: 1928562
- Arevalo J, Ramirez L, Adaui V, Zimic M, Tulliano G, Miranda-Verástegui C, et al. Influence of *Leishmania (Viannia)* species on the response to antimonial treatment in patients with American tegumentary leishmaniasis. J Infect Dis. 2007; 195: 1846–1851. PMID: <u>17492601</u>
- 40. Novais FO, Carvalho LP, Graff JW, Beiting DP, Ruthel G, Roos DS, et al. Cytotoxic T cells mediate pathology and metastasis in cutaneous leishmaniasis. PLoS Pathog. 2013; 9: e1003504. doi: <u>10.1371/journal.ppat.1003504</u> PMID: <u>23874205</u>
- Oliveira DM, Lonardoni MV, Teodoro U, Silveira TG. Comparison of different primers for PCR-based diagnosis of cutaneous leishmaniasis. Braz J Infect Dis. 2011; 15: 204–210. PMID: <u>21670918</u>

- 42. Huse SM, Young VB, Morrison HG, Antonopoulos DA, Kwon J, Dalal S, et al. Comparison of brush and biopsy sampling methods of the ileal pouch for assessment of mucosa-associated microbiota of human subjects. Microbiome. 2014; 2: 5. doi: 10.1186/2049-2618-2-5 PMID: 24529162
- Weina PJ, Neafie RC, Wortmann G, Polhemus M, Aronson NE. Old World leishmaniasis: an emerging infection among deployed US military and civilian workers. Clin Infect Dis. 2004; 39: 1674–1680. PMID: 15578370
- Boggild AK, Valencia BM, Veland N, Pilar Ramos A, Calderon F, Arevalo J, et al. Non-invasive cytology brush PCR diagnostic testing in mucosal leishmaniasis: superior performance to conventional biopsy with histopathology. PLoS One. 2011; 6: e26395. doi: <u>10.1371/journal.pone.0026395</u> PMID: <u>22046280</u>
- 45. Jara M, Valencia BM, Alba M, Adaui V, Arevalo J, Llanos-Cuentas A, et al. Quantitative kDNA assessment during treatment of mucosal leishmaniasis as a potential biomarker of outcome. Poster session presented at: 5<sup>th</sup> World Congress on Leishmaniasis (Worldleish5); 2013 May 13–17; Porto de Galinhas, Pernambuco, Brazil. (Abstract Book, p. 567, abstract nr. 397).
- Brewster S, Barker DC. Analysis of minicircle classes in *Leishmania (Viannia)* species. Trans R Soc Trop Med Hyg. 2002; 96 Suppl 1: S55–63. PMID: <u>12055852</u>
- Prina E, Roux E, Mattei D, Milon G. *Leishmania* DNA is rapidly degraded following parasite death: an analysis by microscopy and real-time PCR. Microbes Infect. 2007; 9: 1307–1315. PMID: 17890124
- **48.** Mary C, Faraut F, Lascombe L, Dumon H. Quantification of *Leishmania infantum* DNA by a real-time PCR assay with high sensitivity. J Clin Microbiol. 2004; 42: 5249–5255. PMID: <u>15528722</u>
- Silveira FT, Lainson R, Corbett CE. Clinical and immunopathological spectrum of American cutaneous leishmaniasis with special reference to the disease in Amazonian Brazil: a review. Mem Inst Oswaldo Cruz. 2004; 99: 239–251. PMID: <u>15273794</u>
- Boggild AK, Ramos AP, Valencia BM, Veland N, Calderon F, Arevalo J, Low DE, Llanos-Cuentas A. Diagnostic performance of filter paper lesion impression PCR for secondarily infected ulcers and nonulcerative lesions caused by cutaneous leishmaniasis. J Clin Microbiol. 2011; 49: 1097–1100. doi: <u>10.</u> <u>1128/JCM.02457-10 PMID: 21177908</u>